Author's response to reviews

Title: A Modified Technique of Single-Incision Laparoscopic Hepaticojejunostomy for Children with Choledochal Cysts

Authors:

Di Xu (cnfjhsh@163.com)

Kunbin Tang (yyu001@icloud.com)

Shaohua HE (slyy2016@126.com;yyu001@me.com)

Version: 3 Date: 27 Mar 2019

Author's response to reviews:

Assistant Editor Comments:

Thank you for clarifying that consent was obtained from a parent or legal guardian. However, we still require a couple of very minor changes.

1. Type of Consent

-- In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Written consent. It is now corrected.

2. Figure List

Please move this after the references.
Done.

3. Clean Manuscript

Done.